Author's response to reviews

Title: A Case of Fungal Keratitis And Onychomycosis Simultaneously Infected By Trichophyton Species

Authors:

Ki Won Jin (kangajis15@gmail.com)
Hyun Sun Jeon (jhs-79@hanmail.net)
Joon Young Hyon (jvhyon@snu.ac.kr)
Wool Suh (being111@hallym.or.kr)
Young Joo Shin (schinn@hanmail.net)

Version: 2
Date: 15 April 2014

Author's response to reviews: see over
Dear Editor

We are enclosing the manuscript entitled “A Case of Fungal Keratitis And Onychomycosis Simultaneously Infected By Trichophyton Species” which I wish to submit for ‘case report’. I would appreciate very much if you could review it for publication.

I had full access to all the data in the study and take responsibility for the integrity of the data and the accuracy of the data analysis as well as the decision to submit for publication. The paper has been written according to the guidelines provided. This manuscript is original material that has not been published or accepted for other publication, in whole or in part, in English or in another language, elsewhere. None of the authors have any financial interests to disclose.

Sincerely,

Correspondence and phone calls about the paper should be directed to me at the following address and phone number:

Young Joo Shin, MD,
Department of Ophthalmology
Hallym University College of Medicine,
948-1 Daerim1-dong, Youngdeungpo-gu, Seoul 150-950,
Korea
Phone: 82-2-829-5193   Fax: 82-2-848-4638
e-mail: schinn@hanmail.net or schinn7@gmail.com